Author’s response to reviews

Title: Robot-assisted laparoscopic reconstructed management of multiple aneurysms in renal artery primary bifurcations: A Case Report and literature review

Authors:

hai-bin wei (haibsjtu@outlook.com)
xiao-long Qi (qixiaolong001@sina.com)
feng liu (liufeng2408@sina.com)
jie wang (542697249@qq.com)
xiao-feng ni (526982806@qq.com)
qi zhang (clinic@126.com)
en-hui li (goodlienhui@163.com)
xuan-yu chen (xuanyuchen@126.com)
da-hong zhang (zhangdahong88@yeah.net)

Version: 2 Date: 23 Aug 2017

Author’s response to reviews:

Dear Prof. Maggie Rahmati and Prof. Hayley Henderson,

Thank you for your letter and advice regarding the review and revision of our manuscript entitled “Robot-assisted laparoscopic reconstructed management of multiple aneurysms in renal artery primary bifurcations: a case report and literature review.” We have addressed all of the comments and have revised the paper again. A point-to-point reply is included as follows. This manuscript has been edited and proofread by EditorBar Language Editing. We appreciate all the reviewers, and thanks for their comments.

We would like to re-submit the revised manuscript for your consideration. We hope that the revision is acceptable, and we look forward to hearing from you soon.
Best Regards!

Yours sincerely,

Hai-bin Wei and Da-hong Zhang

Department of Urology
Zhejiang Provincial People's Hospital, School of Hangzhou Medical College
Email address: haibsjtu@outlook.com and zhangdahong88@yeah.net.

We would like to express our sincere gratitude to the editor for their constructive and positive comments.

Editor Comments:

1) English language:

The standard of English is still not adequate for the manuscript to be accepted. Please ensure that the entire text is further edited by a native/fluent English speaker.

Response: The reviewer’s comment is well appreciated.

Thank you for your reminder. This manuscript has been edited and proofread by EditorBar Language Editing. We expect English language of the manuscript to meet the editorial requirements.

2) Title page:

Please include institutional addresses for all authors.

Response: The reviewer’s comment is well appreciated. We have added the institutional addresses for all authors in the revised manuscript.
3) Declaration section:

* Consent for publication:

We notice that the wording has changed in the revised version of the manuscript. Please change the wording back to that in the original submission, so that the declaration reads as follows:

“We have obtained written informed consent from the patient for publication of this case report and all the accompanying images. A copy of the written consent is available for review by the Editor of the journal.”

Response: The reviewer’s comment is well appreciated. We have changed the declarations section in the revised manuscript.

* Competing interests:

Please amend the current wording to the following:

“The authors declare that they have no competing interests.”

Response: The reviewer’s comment is well appreciated. We have changed the competing interests in the revised manuscript.

4) Tables:

Please move Table 1 to the end of the manuscript, after the References.

Response: The reviewer’s comment is well appreciated. We have moved Table 1 to the end of the manuscript in the revised manuscript. Thank you for your reminder.